

This is the author's manuscript



AperTO - Archivio Istituzionale Open Access dell'Università di Torino

Social media for psychological support of patients with chronic non-infectious diseases: A systematic review

Original Citation:	
Availability:	
This version is available http://hdl.handle.net/2318/1913450	since 2023-06-20T09:48:26Z
Published version:	
DOI:10.1093/heapro/daad006	
Terms of use:	
Open Access Anyone can freely access the full text of works made available as "Open Access". Works made available under a Creative Commons license can be used according to the terms and conditions of said license. Use of all other works requires consent of the right holder (author or publisher) if not exempted from copyright protection by the applicable law.	

(Article begins on next page)

This is a pre-copyedited, author-produced version of an article accepted for publication in Health Promotion International following peer review. The version of record Bert F, Lo Moro G, Giacomini G, Minutiello E, Siliquini R. Social media for psychological support of patients with chronic non-infectious diseases: a systematic review. Health Promot Int. 2023 Apr 1;38(2):daad006 is available online at: 10.1093/heapro/daad006.

Title: Social media for psychological support of patients with chronic noninfectious diseases: a systematic review

Authors:

Fabrizio Bert¹, Giuseppina Lo Moro¹, Gianmarco Giacomini¹, Ettore Minutiello¹, Roberta Siliquini^{1,2}

Affiliations:

Department of Public Health Sciences, University of Turin, Italy

²A.O.U. City of Health and Science of Turin, Italy

*Corresponding author:

Gianmarco Giacomini

Department of Public Health Sciences, University of Turin, Via Santena 5 bis, 10126, Turin, Italy.

Email: gianmarco.giacomini@unito.it

Tel: +39 0116705875

Fax: +39 0116705889

LAY SUMMARY

Psychological symptoms are common among patients with chronic non-infectious diseases. Seldom treated properly, they are a major reason for a low quality of life. Chronic patients, like those who suffer from oncologic, psychiatric or heart diseases, could have difficulties in finding the right psychological support. One way to potentially reach anyone is using social media, which are often used by chronic patients for educational purposes about their diseases, or for finding psychological support from others with similar conditions. We wanted to study whether social media have been used in clinical research for administering psychological support to patients with chronic diseases. We studied the existing scientific literature doing a systematic review, finding that often social media were not used at their full potential. Peerto-peer interactions were the most common interactions, chats rooms and forums the most common social media platforms. Participants were most commonly psychiatric and oncological patients. It is very important not to overlook powerful instruments like social media: they could be easily used by patients to empower themselves in managing their conditions. Social media could be an efficient way to easily provide chronic patients with psychological support either administered by health professionals, like psychotherapists, or by peers.

ABSTRACT

Psychological complications of chronic diseases are often underestimated because they may be interpreted as normal consequences of the underlying illness. Additionally, chronic patients may find several obstacles to healthcare access, including physical, socioeconomic, geographical, and psychological barriers. Social media may be potential tools to provide psychological care extending the possibility of treatment where the offer is limited. This review aimed to explore the use of social media in administering psychological interventions to patients with chronic noncommunicable diseases. Following the PRISMA guidelines, a systematic review was conducted by searching PubMed, Embase, and PsycInfo (from 2004 to 2021). Randomized Controlled Trials outlining the social media use in administering psychological assistance to patients with chronic noncommunicable diseases were considered eligible. Out of 9838 records identified, 75 papers were included. Peer-to-peer interaction was the most used (n=22), mainly via chat rooms and forums. Interventions were mostly administered to patients with psychiatric (n=41) or oncologic disorders (n=12). This work highlighted a lack of tailored interventions based on disease, age, or gender, and a use of a limited range of relatively old platforms such as emails, blogs, and forums. To administer efficient interventions, it would be advisable to continuously analyze the evolving use of

Breakdown of the total word count:

Lay summary: 198

these tools.

Abstract: 200

Background: 580

Methods: 1001

Results: 1607

Discussion: 1112

Conclusion: 161

BACKGROUND

Having a chronic illness is known to be associated with an increased risk for psychological disorders (Daré et al., 2019; Ducat et al., 2014). Even so, the psychological dimension of people living with chronic illness is often underestimated or neglected (WHO, 2018; Voinov et al., 2013).

The link between chronic illnesses and psychological disorders is even more relevant considering that 13 out of the 20 diseases that caused most of disability-adjusted life years (DALYs) in 2019 were chronic diseases, such as ischemic heart disease, stroke, chronic obstructive pulmonary disease, diabetes, depression, and others (Bochen Cao et al., 2020). Moreover, 41 million people die every year from a chronic disease, accounting for over 70% of all deaths worldwide (Riley L et al., 2020).

The mistaken belief that some symptoms (e.g. sleep disturbance, fatigue, lack of appetite) are an expected consequence of the underlying chronic condition can bring to the underdiagnosis of psychological complications, leading to a worsening of the treatment outcome (Osservatorio Nazionale sulla salute nelle regioni italiane, 2019; Voinov et al., 2013). Considering the existence of several barriers to access healthcare for chronic patients (such as physical, socioeconomic, geographical, and psychological barriers) (Foo et al., 2020; Stein et al., 2019), having psychological comorbidity while suffering from a chronic disease could even complicate finding access to support care (Stein et al., 2019). Moreover, social stigma goes with psychological disorders, making it more difficult to find support and affecting its quality (Baumann & Dang, 2012).

Given that around 4.2 billion people use social media globally (Starri, 2021), these could be potentially appropriate tools to provide psychological support to the chronically ill from all

social extractions, extending the possibility of treatment to lower social classes or to those living in rural areas where the offered treatments are limited (Scogin et al., 2018). Moreover, it should be noted that many patients suffering from chronic diseases use the internet or social networks to search for useful information about their condition (Fox S., 2011). Social media could meet some needs that may not be detected during follow-up, as it is shown by the fact that many patients use social media to connect with other chronic patients in forums, blogs or chats to raise awareness of the disease (Fox S., 2011).

The role of social media in healthcare contexts has been investigated several times, for example for raising awareness about the prevention and care of HIV (Latkin et al., 2013; Muessig et al., 2015; Ronen et al., 2020), for health promotion and communication purposes (Smailhodzic et al., 2016; Chou 2013), for lifestyle modification and diabetes management (Cotter, 2014) or for administering mental health interventions or psychiatric treatment (Kazdin, 2019; Alvarez-Jimenez M., 2014; Naslund, 2016), especially in adolescents (Keles et al. 2020; Best et al., 2014). However, currently, too little has been studied about whether and how social media are used in robust clinical research, specifically in randomized controlled trials (RCTs), for psychological support in patients with chronic non-communicable diseases.

Therefore, this review aims to investigate which social media are commonly used in RCT studies, understand the preferred features and point out in which nosological categories and for what kind of psychological complication these tools are mostly used. This could provide new insights into social media's role in improving fragile populations' mental health.

Updating knowledge about the current state of the art of social media usage in clinical research is pivotal to pave the way for further research for an ever-improving application of this potentially promising instrument.

METHODS

Data sources and search strategy

A systematic review was conducted following the Preferred Reporting Items for Systematic Reviews and Meta-Analyses (PRISMA) checklist (Page et al., 2021) by searching PubMed, Embase, and PsycInfo databases. As suggested by the Cochrane handbook for systematic reviews (Lefebvre et al., 2022), PubMed and Embase were chosen as generalist bibliographic databases, while PsycInfo was chosen as specialist bibliographic database to be used when studying psychological interventions.

The search was carried out on March 25th, 2021. Dissertations were filtered out (PsycInfo). The authors selected papers published from January 1st, 2004, through March 25th, 2021. The starting point from 2004 was chosen because MySpace and Facebook, the first popular social networks, were launched in 2003 and 2004, respectively, thus contributing in coining the term "Social Media" and promoting the importance of social networks as we know it today (A. M. Kaplan & Haenlein, 2009). In addition, the essential concepts for the definition of "Social Media" (i.e. Web 2.0 and User Generated Content, UGC) were developed in 2004. Indeed, we decided to adopt the definition given by Kaplan and Haenlein "Social Media is a group of Internet-based applications that build on the ideological and technological foundations of Web 2.0, and that allow the creation and exchange of UGC" (A. M. Kaplan & Haenlein, 2009). Summing up, the Web 2.0 is a platform where users repeatedly participate and collaborate to change content and applications and the UGC represents the various ways of media content that are created by users (A. M. Kaplan & Haenlein, 2009). Similar

definitions of social media are commonly shared and the interactivity between users is among the main key characteristics (Cao et al., 2017; Obar & Wildman, 2015).

Both free terms and thesaurus terms were used to create the search strategy. Search terms for four main themes were used: chronic diseases AND psychological care AND social media AND randomized controlled trial. Supplementary file 1 shows the full search strings. The field of chronic conditions is extremely wide and includes noncommunicable conditions, long-term mental disorders, ongoing physical/structural impairments and persistent communicable conditions (WHO, 2002). To circumscribe our search, we decided to exclude chronic conditions related to communicable diseases as other reviews have already focused on social media and some of these conditions (Cao et al., 2017; Ronen et al., 2020). Based on this, in order to find papers about the most relevant chronic conditions, we decided to consider the top 35 noncommunicable disease causes of disability-adjusted life years (DALYs) among people aged above 15 years (Bochen Cao et al., 2020) (including mental disorders and impairments). Overall, these top causes included cardiovascular diseases, cancers, diabetes mellitus, pain conditions, respiratory diseases, neurological diseases, mental disorders, digestive diseases, hearing loss, gynecological diseases, musculoskeletal disorders, endocrine, blood and immune disorders, refractive errors, kidney diseases, and edentulism. Both synonyms of "chronic diseases" and terms referring to the above-mentioned conditions were included in the search string.

About terms for social media, we used the definition and classification of Kaplan and Haelein, which include blogs, social networking sites, virtual social worlds, collaborative projects, content communities, and virtual game worlds (A. M. Kaplan & Haenlein, 2009). Both common nouns and proper nouns of the most used social media were included in the search string.

Last, we decided to focus on RCTs since we were interested in exploring how social media were used compared with other traditional supports to eventually analyze in future papers the efficacy of social media interventions with the best grade of evidence.

Study selection and data extraction

Table 1 shows the PICOS (Population, Intervention, Comparator, Outcome, Study type) approach (Munn et al., 2018) that guided the inclusion criteria along with the abovementioned definitions of chronic conditions and social media (A. M. Kaplan & Haenlein, 2009; WHO, 2002). Specifically, studies were considered eligible if their sample was composed of people with chronic conditions (noncommunicable conditions, long-term mental disorders, ongoing physical/structural impairments (WHO, 2002), including but not limited to the top 35 noncommunicable disease causes of DALYs (Bochen Cao et al., 2020). To be eligible, the interventions must include any kind of psychological care provided through social media, i.e. media that had both these two essential features: internet-based platform and exchange of any kind of content between users.

[insert - Table 1. PICOS approach - here]

No restrictions were applied regarding the users with whom participants interacted (e.g. peers, therapists, physicians, family members). Only peer-reviewed studies with RCT design and papers written in English, French, Spanish, Italian, or Portuguese were considered eligible.

Studies with samples consisting exclusively of people below a threshold for a disorder (e.g. subthreshold depression), of patients with communicable diseases, and of caregiver/relatives of patients were excluded. Papers with interventions with no interactions between users or based exclusively on chatbot/automated virtual assistant and smoking cessations interventions were excluded. Protocols, letters, editorials, commentaries, and reviews were excluded.

In a first phase, duplicates were removed, and articles were independently screened by the authors (EM, GG, GLM) for title and abstract via the Rayyan software (Ouzzani et al., 2016). Disagreements were solved by a consensus-based discussion. In a second phase, the authors (EM, GG, GLM) independently applied the inclusion and exclusion criteria to the full texts. Disagreements were resolved by consensus and reasons for exclusion were documented. Then, the authors (EM, GG, GLM) extracted the following information from the final included studies: year of publication, country, language, RCT features, participants and setting, length of the intervention, year of study, outcome measures, funding and conflict of interest information, intervention and control group characteristics. The last two fields were analyzed to assess which social media platforms were employed in the intervention or control group, which were the interactions between participants (or between participants and study operators), and which roles social media had in the intervention. The above-mentioned roles were categorized based on common characteristics, to summarize their application. The authors carried out a narrative synthesis according to study design, setting, population included, outcome assessed, intervention or control design and characteristics. In addition, the extracted data are presented in Summary of Findings tables.

RESULTS

A total of 9838 records (Pubmed: 4896; Embase: 2957; PsycInfo: 1985) were identified through database searching. Figure 1 represents an overview of the study selection process following the Preferred Reporting Items for Systematic Reviews and Meta-Analyses (PRISMA) flow diagram (PRISMA 2020). The whole selection phase resulted in 75 included publications (Ahola Kohut et al., 2016; Alavi et al., 2016; Anastasiadou et al., 2020; Andersson et al., 2012; Arnold et al., 2019; Aubin et al., 2018; Bogosian et al., 2015, 2022; Bostock et al., 2016; Boyd et al., 2019; Buhrman et al., 2013; Chavooshi, Mohammadkhani, & Dolatshahee, 2016; Chavooshi, Mohammadkhani, & Dolatshahi, 2016; Chee et al., 2016; H. Chen et al., 2020; Choi et al., 2014; Chong & Moreno, 2012; Classen et al., 2013; Conrad et al., 2015; Glass et al., 2017; Gliddon et al., 2019; Graham et al., 2020; Greenberg et al., 2019; Heim et al., 2021; Hensel, [..] Bouck, et al., 2019; Hensel, [..] Cohen, et al., 2019; Hesser et al., 2012; Hoffmann et al., 2021; Hur et al., 2018; Jasper et al., 2014; Johansson et al., 2019; Kim et al., 2018; Kryger et al., 2019; Krzystanek et al., 2019; Li & Tse, 2020; Linden et al., 2018; Ljótsson et al., 2010; Lundgren et al., 2016; Ly et al., 2014; Ma et al., 2021; Moberg et al., 2019; Mohr et al., 2019; Neumayr et al., 2019; Nolte et al., 2021; Omidi et al., 2020; Oser et al., 2019; Owen et al., 2017; Pan et al., 2020; Perini et al., 2009; Robinson et al., 2010; Robinson-Whelen et al., 2020; Ruehlman et al., 2012; Ruland et al., 2013; Schlosser et al., 2018; Schulz et al., 2020; Scogin et al., 2018; Stinson et al., 2016; Strandskov et al., 2017; Titov, Andrews, Choi, et al., 2008; Titov, Andrews, Davies, et al., 2010; Titov, Andrews, Johnston, et al., 2010; Titov, Andrews, Schwencke, et al., 2008; Titov et al., 2009, 2011; Topooco et al., 2019; Vilhauer et al., 2010; Wang et al., 2018; Weise et al., 2016, 2019; Wims et al., 2010; Wise et al., 2018; Wootten et al., 2015, 2017; Young et al., 2018; Zwerenz et al., 2017).

Characteristics of the included studies

The included papers were published between 2008 and 2021, and more than half of them (n=44) were written after 2017. The year with more articles (n=14) was 2019.

Most of the studies presented data from the United States of America (n=19), Australia (n=13) and Sweden (n=10) and they were mostly designed with two arms (n=62): an intervention and a control group. However, 12 studies were designed with three arms. In 19 studies control groups consisted of a waiting list, in 14 studies they consisted of a discussion forum group, and in 13 studies they received educational material. In 14 studies the control group received a deferred treatment.

The sample size ranged from 21 to 1013 participants (Supplemental Table S1), the ratio of intervention participants to control participants ranged from 1:1 to 2.7:1. The length of the study ranged from three weeks to 60 weeks; eight studies had a follow up period - lasting one month up to 12 months.

Seven studies were feasibility RCTs and two were pilot RCTs. A total of 4 studies were multicenter RCTs. Fifty-nine works had a declared funding source and 13 studies stated potential conflicts of interest. All studies were written in English. Full details of characteristics of the included studies are presented in Supplemental Table S1.

Characteristics of participants and chronic diseases

The mean age of participants ranged from 15.5 years to 65 years. Females represented the majority of respondents in 46 studies and the totality of respondents in 10 studies, whereas males represented the majority in 16 studies and the totality in 2 studies.

Most patients were recruited in clinics/hospitals (n=27), with a mean age of 15-58 years, via social media such as Facebook or Forums (n=9), with a mean age of 17-47 years, or via websites/advertising online (n=11), with mean age 38-61 years. Eighteen studies recruited patients via multiple methods (Supplemental Table S2). Some inclusion criteria were frequent across the included studies, such as being at least 18 years old (n=45) and having access to the internet (n=42). The most frequent exclusion criteria for participants were severe psychiatric condition or suicidality (n=47), pregnancy (n=4), severe cognitive impairment (n=7), and addictive disorders (n=7) (Supplemental Table S3).

Most of the studies concerned psychological support in psychiatry (n=41) and in oncology (n=12). The studies in the psychiatric area were mainly about depressive disorders (n=10), anxiety disorders (n=7) or both (n=4). The others were mostly about phobias (n=2) and eating disorders (n=2). The studies in the oncology area were mostly about breast (n=4), prostate (n=2) or gynecologic cancer (n=1); the others were about cancer survivors or patients with cancer in general (Supplemental Table S2).

Considering the diseases addressed in both genders, the studies mostly concerned psychiatric and oncological areas. Studies with a higher percentage of males mainly addressed cardiological (n=3) and otolaryngological (n=3) diseases; whereas studies with a higher percentage of females mostly addressed chronic pain (n=4) and neurological (n=4) diseases.

Characteristics of the interventions and controls

The main characteristics of interventions and controls are reported in Table 2 and Supplemental Table S4, respectively. Thirty-six studies administered interventions via ad hoc social media tools (forums or in app messaging), while two via pre-existing social media tools. Fourteen studies administered interventions either via both ad hoc tools and pre-

existing ones. Seventeen studies have had more than one type of intervention, as explained below.

Regarding ad hoc tools, chats have mostly been used (n=15), then forums (n=14). One study used social media as a blog, one as a social game, one for web calls. As for the existing tools, Skype was the most used (n=10), then emailing systems (n=7), WeChat (n=4), and Facebook (n=2). One study used Telegram, one SecondLife. The studies that administered the interventions with two social media used mostly forums plus emails or chats or, in one case, emails and a community.

In 21 studies social media tools were used in the control group too. Fourteen studies administered control via forum. Facebook, Skype, chatting, and SecondLife were used once. Only one study administered control in a mixed way (via email and chat).

The characteristics of the interaction between users changed based on the features of the platform that was being used. In 22 studies the intervention was driven by a peer-to-peer interaction with or without a supervisor coach - most often a trained peer. Among these studies, some administered interventions via chat or social games, but also web calls and forums. In ten studies patients interacted with coaches only, in 19 studies patients interacted with psychologists or psychiatrists, while in four studies patients interacted with nurses or clinicians. In 20 studies the interaction was mixed. Overall, peer to peer interactions were the most recurrent.

In the studies that provided psychological support to patients with psychiatric conditions, nearly all participants were under 50 years (n=37) and most studies had a higher percentage of females (n=35 out of 41). Instead, in the other pathological areas, patients were more likely between their 40s and 60s (n=28), and studies with a higher percentage of males were more represented (n=12 out of 34). Apart from the demographic differences, in both psychiatric and non-psychiatric conditions the interventions were substantially administered by the same

platforms and without any outstanding peculiarities: mostly forums, chats and web-calling platforms were used.

Based on their features, social media platforms were used with different interactions and purposes. Chats and emails were frequently used for receiving feedbacks about how the intervention proceeded, for receiving educational material or additional support other than the intervention per se. This was more frequent for non-peer to peer interactions, like therapist/health professional-driven or coach-driven interventions. On the other hand, forums and communities were more frequently used for establishing interactions between users, having a peer-to-peer interplay. Web-calling tools were used for administering telepsychotherapy.

Regarding the above-mentioned purposes, the most recurring ones were the "sharing emotions or feelings" - which went along with forums and chats, and it was mostly common in the "psychiatric area" studies, but largely diffused among all others - and the "feedback" one, which went along with chats and emails. In some studies, social media were used with more than one purpose, often mixing the "sharing emotions or feelings" one with the "feedback" one.

Psychiatric and oncological studies - the most recurring areas of interest in the evaluated studies - were more diversified than all other conditions, in terms of which and how social media platforms were used.

[insert - Table 2. Characteristics of the interventions - here]

As for the controls administered via social media, given that they were administered eminently via forum, the role of social media was almost only the "sharing emotions or feelings" one.

Outcome measures

Most outcomes were evaluated via questionnaires or validated scales that mostly were: Patient Health Questionnaire-9 (PHQ-9) or other versions of it (n=24); General Anxiety Disorder-7 (GAD-7) or other versions (n=14); Hospital Anxiety and Depression Scale (HADS) (n=12); Sheehan Disability Scale (SDS) (n=11); Beck Depression Inventory®-II (BDI-II) or other versions (n=9). All studies had more than one outcome to measure, and they used more than one scale or questionnaire to evaluate patients' depression, anxiety, quality of life, functional impairment, and pain.

Among the studies with social media driven interventions, 53 studies reported significant improvements in patients' psychological outcomes; 20 reported no significant results; 2 stated to have obtained insufficient data due to the insufficient sample size of their studies. None of the studies reported a negative effect (Supplemental Table S5).

DISCUSSION

This review highlighted the use of social media to provide psychological support for patients with chronic non-communicable diseases. Overall, 75 RCTs published between 2008 and 2021 were included. Most patients were women with an age range from 15.5 to 65, affected by psychiatric or oncologic diseases. Interventions were mostly administered as peer-to-peer interactions via chat rooms and forums, allowing the participants to share their feelings. Additionally, interactions via web-calling platforms, chat rooms or forums with psychotherapists or health professionals were common too, allowing participants to receive psychotherapy or feedback.

No outstanding differences in social media usage between different settings (such as different diseases or different patients' gender and age) were identified.

There might be some possible explanations for the broad involvement of psychiatric and oncologic patients. Social media, thanks to peer-to-peer interaction, are already known for their usefulness in mental health care (Naslund et al., 2016). Moreover, psychiatric patients have needs that are seldom adequately addressed, requiring a wide choice of treatment options (Lake & Turner, 2017) that may result in the necessity to study new ways to administer psychological interventions. Furthermore, oncologic patients are well known for their needed psychological support, although too often their psychological symptoms are confused as an expected consequence of cancer disease (Merckaert et al., 2010), underestimating the occurrence of psychological complications in oncologic patients (Grassi et al., 2017).

The extensive use of peer-to-peer interaction is in line with previous research (Betton et al, 2015; Chen & Wang, 2021). Indeed, the effectiveness of peer-to-peer support groups is already known for many conditions (Evans et al., 2021). They provide emotional, appraisal and informational assistance shared with members suffering from the same condition (Doull et al., 2017). Moreover, peers could give importance to aspects that might be overlooked by health professionals, such as hearing about others' experiences undergoing specific treatments, and performing self-care activities. (Rupert et al., 2016). Participants can document and share their progress, setting common goals, supporting, motivating and educating each other, even when moderated by health professionals to avoid misinformation and the spreading of fake news (Chen & Wang, 2021). Peer-to-peer interaction has proven to be effective also in engaging those hard-to-reach populations, not only geographically

speaking, but also those who fear reaching out for help (Prescott et al., 2020; Sokol & Fisher, 2016).

In this perspective, social media that allow peer-to-peer interaction could reduce social stigma and give a sense of belonging to a community (Smailhodzic et al., 2016; Naslund et al., 2016). The possibility of not revealing one's identity, the asynchronicity of interactions, the possibility of getting support 24/7, and the chance to talk with someone who shares similar experiences and difficulties are all relevant aspects in challenging the stigma and the fear of being disapproved or marginalized for one's diagnosis (Chen & Wang, 2021; Naslund et al., 2016).

During interventions, health professionals mainly provided feedback on progress or at most some instructions. These kinds of interventions, thanks to better follow-up execution and better teaching about disease management, are reported to increase psychological well-being and self-efficacy levels (Smailhodzic et al., 2016; Wang et al., 2018; Karabulut et al., 2014; Pendergast et al., 2017; Fukuoka et al., 2015). Furthermore, most interventions involving interaction with psychotherapists aimed to administer telepsychotherapy, a well-known practice since the 90s (E. H. Kaplan, 1997). This takes advantage of social media platforms like Skype to deliver videoconferencing even if not without important hindrances, such as reticence of professionals, lack of digital literacy and training of both professionals and patients (Van Daele et al., 2020).

Surprisingly, social media were used quite similarly in all contexts, despite the heterogeneity of patients and diseases involved. Additionally, only a few kinds of social media platforms were used, mostly earlier ones like emails, blogs, and forums. However, some variability in the kind and usage of platforms emerged in the two main nosology categories (psychiatry and

oncology) (Table 2). These findings may suggest that the choice in social media platforms is driven by usability, rather than by their characteristics or those of the target population (like gender, age, or disease). Taking into account that outcomes could change depending on programs used in interventions, it could be helpful to tailor social media to fit the patient's needs, considering how different populations experience social media (e.g. differences among generations) (Pew Research Center, 2021; Smith & Anderson, 2018).

Most interventions were about health promotion, focusing on empowering chronic sufferers about: living with their disease, managing their symptoms, and increasing awareness of their condition. Social media are easily accessible from personal computers or smartphones, opening up the possibility of the patient's home becoming the first place of care. (Smailhodzic et al., 2016; Stellefson et al., 2013). Thus, health promotion via social media could help improve the main activities of domiciliary care, among which there are both empowerment and psychological support (Cirio et al., 2021; Genet et al., 2011).

According to the affordances theory (Merolli et al., 2013), features offered to users are more important than the social media itself. For instance, among the affordances of social media assessed in our review, the possibility of anonymity helps sufferers to speak without fear of being judged, while asynchronous communication and flexibility allow participants to interact whenever they want. Additionally, social media are in constant evolution, adapting to users' needs.

In conclusion, social media could extend psychological care outside of healthcare facilities.

The two most common interactions (peer-to-peer and psychotherapist) are both effective supports that are normally provided in health facilities for treating psychological issues.

Administering them using social media might put patients at the center, providing a time and

space extension of routine care (Mohr et al., 2015). Considering the current decrease in human and financial resources faced by health facilities and, on the other hand, the increasing demand for a more holistic and patient-centered approach, social media could be an inexpensive, easy-to-use, and extremely diffused tool to face those issues.

Limitations

Regarding limitations, we decided to focus on the top 35 noncommunicable diseases causing DALYs in people over 15 years of age. Possibly, by extending the research to other chronic noncommunicable diseases and expanding the age range of the subjects, the results could change. Also, this study does not consider chronic conditions related to communicable diseases. The search was carried out using three databases, so the results could change by extending the search to other databases. Furthermore, scientific literature may be limited in showing the real-world use of apps or social media, which can be offered to the population without an evidence-based study of efficacy (Bert et al., 2016). Finally, given the heterogeneity of the results, we did not conduct a meta-analysis to assess whether the interventions were effective or not.

CONCLUSION

This work contributes to shed some light on the actual social media usage in clinical research, describing the characteristics of social media-driven interventions for psychological support in patients with chronic diseases and their potential application in future tailored interventions on the specific patients' needs.

In clinical research, it seemed that social media were not used to their full potential, chosen mainly by usability, rather than by their characteristics or those of the target population.

Therefore, it will be essential to analyze how these tools evolve, understanding the complex interactions between participants to administer more efficient interventions, directly into the patient's house.

In conclusion, from a public health perspective, it will be fundamental for policymakers to consider the potential of social media for the implementation of telemedicine. This is relevant especially considering the COVID-19 pandemic, which highlighted the need for new approaches to patients that can quickly and effectively reach more people, overcoming physical, psychological, and geographic barriers to accessing care.

Funding details

None.

Disclosure statement

The authors report there are no competing interests to declare.

Data availability statement

All relevant data are within the paper.

Ethics information

Not applicable.

Acknowledgments

Not applicable.

REFERENCES

Ahola Kohut, S., Stinson, J. N., Ruskin, D., Forgeron, P., Harris, L., van Wyk, M., Luca, S., & Campbell, F. (2016). iPeer2Peer program: a pilot feasibility study in adolescents with chronic pain. PAIN, 157(5).

https://journals.lww.com/pain/Fulltext/2016/05000/iPeer2Peer_program__a_pilot_feasibility study in.20.aspx

Alavi, N., Hirji, A., Sutton, C., & Naeem, F. (2016). Online CBT Is Effective in Overcoming Cultural and Language Barriers in Patients With Depression. Journal of Psychiatric Practice®, 22(1).

 $https://journals.lww.com/practicalpsychiatry/Fulltext/2016/01000/Online_CBT_Is_Effective_in_Overcoming_Cultural_and.2.aspx$

Alvarez-Jimenez, M et al. Online, social media and mobile technologies for psychosis treatment: a systematic review on novel user-led interventions. Schizophrenia research vol. 156,1 (2014): 96-106. doi:10.1016/j.schres.2014.03.021

Anastasiadou, D., Folkvord, F., Brugnera, A., Cañas Vinader, L., SerranoTroncoso, E., Carretero Jardí, C., Linares Bertolin, R., Muñoz Rodríguez, R., Martínez Nuñez, B., Graell Berna, M., Torralbas-Ortega, J., Torrent-Solà, L., Puntí-Vidal, J., Carrera Ferrer, M., Muñoz Domenjó, A., Diaz Marsa, M., Gunnard, K., Cusido, J., Arcal Cunillera, J., & Lupiañez-Villanueva, F. (2020). An mHealth intervention for the treatment of patients with an eating

disorder: A multicenter randomized controlled trial. International Journal of Eating Disorders,

53(7), 1120–1131. https://doi.org/https://doi.org/10.1002/eat.23286

Andersson, G., Carlbring, P., Furmark, T., & Group, S. O. F. I. E. R. (2012). Therapist experience and knowledge acquisition in internet-delivered CBT for social anxiety disorder: a randomized controlled trial. PloS One, 7(5), e37411–e37411.

https://doi.org/10.1371/journal.pone.0037411

Arnold, C., Villagonzalo, K.-A., Meyer, D., Farhall, J., Foley, F., Kyrios, M., & Thomas, N. (2019). Predicting engagement with an online psychosocial intervention for psychosis:

Exploring individual- and intervention-level predictors. Internet Interventions, 18.

https://doi.org/10.1016/j.invent.2019.100266

Aubin, S., Rosberger, Z., Hafez, N., Noory, M. R., Perez, S., Lehmann, S., Batist, G., &

Kavan, P. (2018). Cancer!? I Don't Have Time for That: Impact of a Psychosocial

Intervention for Young Adults with Cancer. Journal of Adolescent and Young Adult

Oncology, 8(2), 172–189. https://doi.org/10.1089/jayao.2017.0101

Baumann, L. C., & Dang, T. T. N. (2012). Helping patients with chronic conditions overcome barriers to self-care. Nurse Practitioner, 37(3), 32–38.

https://doi.org/10.1097/01.NPR.0000411104.12617.64

Bert, F., Passi, S., Scaioli, G., Gualano, M. R., & Siliquini, R. (2016). There comes a baby! What should I do? Smartphones' pregnancy-related applications: A web-based overview. Health Informatics Journal, 22(3), 608–617. https://doi.org/10.1177/1460458215574120 Best, P., Manktelow, R., & Taylor, B. (2014). Online communication, social media and adolescent wellbeing: A systematic narrative review. Children and Youth Services Review, 41, 27–36. https://doi.org/10.1016/j.childyouth.2014.03.001

Betton, V., Borschmann, R., Docherty, M., Coleman, S., Brown, M., & Henderson, C. (2015). The role of social media in reducing stigma and discrimination. The British Journal of

Psychiatry, 206(6), 443-444. https://doi.org/10.1192/BJP.BP.114.152835

Bochen Cao, by, Stevens, G. A., Ho, J., Ma Fat, D., Cao Gretchen Stevens, B. A., Ma, D., Estevez, D., Daher, J., Ferlay, J., Gacic-Dobo, M., Gerland, P., Glaziou, P., Hug, L., Iaych, K., Jakob, R., Liu, L., Lozano, R., Mahy, M., Mathers, C., ... You, D. (2020). WHO methods and data sources for country-level causes of death 2000-2019.

https://cdn.who.int/media/docs/default-source/gho-documents/global-health-estimates/ghe2019 cod methods.pdf

Bogosian, A., Chadwick, P., Windgassen, S., Norton, S., McCrone, P., Mosweu, I., Silber, E., & Moss-Morris, R. (2015). Distress improves after mindfulness training for progressive MS: A pilot randomised trial. Multiple Sclerosis Journal, 21(9), 1184–1194.

https://doi.org/10.1177/1352458515576261

Bogosian, A., Hurt, C. S., Hindle, J. v, McCracken, L. M., Vasconcelos E Sa, D. A., Axell, S., Tapper, K., Stevens, J., Hirani, P. S., Salhab, M., Ye, W., & Cubi-Molla, P. (2022). Acceptability and Feasibility of a Mindfulness Intervention Delivered via Videoconferencing for People With Parkinson's. Journal of Geriatric Psychiatry and Neurology, 35(1), 155–167. https://doi.org/10.1177/0891988720988901

Bostock, S., Luik, A. I., & Espie, C. A. (2016). Sleep and Productivity Benefits of Digital Cognitive Behavioral Therapy for Insomnia: A Randomized Controlled Trial Conducted in the Workplace Environment. Journal of Occupational and Environmental Medicine, 58(7). https://journals.lww.com/joem/Fulltext/2016/07000/Sleep_and_Productivity_Benefits_of_Digital.6.aspx

Boyd, R. C., Price, J., Mogul, M., Yates, T., & Guevara, J. P. (2019). Pilot RCT of a social media parenting intervention for postpartum mothers with depression symptoms. Journal of

Reproductive and Infant Psychology, 37(3), 290–301.

https://doi.org/10.1080/02646838.2018.1556788

Buhrman, M., Fredriksson, A., Edström, G., Shafiei, D., Tärnqvist, C., Ljótsson, B., Hursti, T., Gordh, T., & Andersson, G. (2013). Guided Internet-delivered cognitive behavioural therapy for chronic pain patients who have residual symptoms after rehabilitation treatment: Randomized controlled trial. European Journal of Pain, 17(5), 753–765.

https://doi.org/https://doi.org/10.1002/j.1532-2149.2012.00244.x

Cao, B., Gupta, S., Wang, J., Hightow-Weidman, L. B., Muessig, K. E., Tang, W., Pan, S., Pendse, R., & Tucker, J. D. (2017). Social Media Interventions to Promote HIV Testing, Linkage, Adherence, and Retention: Systematic Review and Meta-Analysis. Journal of Medical Internet Research, 19(11). https://doi.org/10.2196/JMIR.7997

Chavooshi, B., Mohammadkhani, P., & Dolatshahee, B. (2016). Telemedicine vs. in-person delivery of intensive short-term dynamic psychotherapy for patients with medically unexplained pain: A 12-month randomized, controlled trial. Journal of Telemedicine and Telecare, 23(1), 133–141. https://doi.org/10.1177/1357633X15627382

Chavooshi, B., Mohammadkhani, P., & Dolatshahi, B. (2016). A Randomized Double-Blind Controlled Trial Comparing Davanloo Intensive Short-Term Dynamic Psychotherapy as Internet-Delivered Vs Treatment as Usual for Medically Unexplained Pain: A 6-Month Pilot Study. Psychosomatics, 57(3), 292–300.

https://doi.org/https://doi.org/10.1016/j.psym.2016.01.001

Chee, W., Lee, Y., Im, E.-O., Chee, E., Tsai, H.-M., Nishigaki, M., Yeo, S. A., Schapira, M. M., & Mao, J. J. (2016). A culturally tailored Internet cancer support group for Asian American breast cancer survivors: A randomized controlled pilot intervention study. Journal of Telemedicine and Telecare, 23(6), 618–626. https://doi.org/10.1177/1357633X16658369

Chen, H., Cai, C., & Xie, J. (2020). The effect of an intensive patients' education program on anxiety, depression and patient global assessment in diabetic foot ulcer patients with Wagner grade 1/2: A randomized, controlled study. Medicine, 99(6), e18480–e18480.

Chen, J., & Wang, Y. (2021). Social Media Use for Health Purposes: Systematic Review. Journal of Medical Internet Research, 23(5). https://doi.org/10.2196/17917

Choi, N. G., Hegel, M. T., Marti, N., Marinucci, M. L., Sirrianni, L., & Bruce, M. L. (2014). Telehealth problem-solving therapy for depressed low-income homebound older adults. The American Journal of Geriatric Psychiatry: Official Journal of the American Association for Geriatric Psychiatry, 22(3), 263–271. https://doi.org/10.1097/JGP.0b013e318266b356 Chong, J., & Moreno, F. (2012). Feasibility and Acceptability of Clinic-Based Telepsychiatry for Low-Income Hispanic Primary Care Patients. Telemedicine and E-Health, 18(4), 297–304. https://doi.org/10.1089/tmj.2011.0126

Chou, Wen-ying Sylvia et al. Web 2.0 for health promotion: reviewing the current evidence. American journal of public health vol. 103,1 (2013): e9-18.

doi:10.2105/AJPH.2012.301071

https://doi.org/10.1097/MD.000000000018480

Cirio, F., Pontillo, C., & Zanchi, F. (2021, December 10). Le cure domiciliari alla popolazione della città di Torino | Articolo | welforum.it. https://welforum.it/le-cure-domiciliari-alla-popolazione-della-citta-di-torino/

Classen, C. C., Chivers, M. L., Urowitz, S., Barbera, L., Wiljer, D., O'Rinn, S., & Ferguson, S. E. (2013). Psychosexual distress in women with gynecologic cancer: a feasibility study of an online support group. Psycho-Oncology, 22(4), 930–935.

https://doi.org/https://doi.org/10.1002/pon.3058

Conrad, I., Kleinstäuber, M., Jasper, K., Hiller, W., Andersson, G., & Weise, C. (2015). The Changeability and Predictive Value of Dysfunctional Cognitions in Cognitive Behavior

Therapy for Chronic Tinnitus. International Journal of Behavioral Medicine, 22(2), 239–250. https://doi.org/10.1007/s12529-014-9425-3

Cotter, Alexander P et al. Internet interventions to support lifestyle modification for diabetes management: a systematic review of the evidence. Journal of diabetes and its complications vol. 28,2 (2014): 243-51. doi:10.1016/j.jdiacomp.2013.07.003

Daré, L. O., Bruand, P. E., Gérard, D., Marin, B., Lameyre, V., Boumédiène, F., & Preux, P. M. (2019). Co-morbidities of mental disorders and chronic physical diseases in developing and emerging countries: A meta-analysis. BMC Public Health, 19(1), 1–12. https://doi.org/10.1186/S12889-019-6623-6/TABLES/1

Doull, M., O'Connor, A. M., Welch, V., Tugwell, P., & Wells, G. A. (2017). Peer support strategies for improving the health and well-being of individuals with chronic diseases. The Cochrane Database of Systematic Reviews, 2017(6).

https://doi.org/10.1002/14651858.CD005352.PUB2

Ducat, L., Philipson, L. H., & Anderson, B. J. (2014). The Mental Health Comorbidities of Diabetes. JAMA, 312(7), 691. https://doi.org/10.1001/JAMA.2014.8040

Evans, M., Daaleman, T., & Fisher, E. B. (2021). Peer Support for Chronic Medical Conditions. Peer Support in Medicine, 49–69. https://doi.org/10.1007/978-3-030-58660-7_3 Foo, K. M., Sundram, M., & Legido-Quigley, H. (2020). Facilitators and barriers of managing patients with multiple chronic conditions in the community: a qualitative study.

BMC Public Health, 20(1). https://doi.org/10.1186/S12889-020-8375-8

Fox S. (2011). The Social Life of Health Information, 2011 |.

https://www.pewresearch.org/internet/2011/05/12/the-social-life-of-health-information-2011/

Fukuoka, Yoshimi et al. "A Novel Diabetes Prevention Intervention Using a Mobile App: A Randomized Controlled Trial With Overweight Adults at Risk." American journal of preventive medicine vol. 49,2 (2015): 223-37. doi:10.1016/j.amepre.2015.01.003

Genet, N., Boerma, W. G., Kringos, D. S., Bouman, A., Francke, A. L., Fagerström, C., Melchiorre, M., Greco, C., & Devillé, W. (2011). Home care in Europe: A systematic literature review. BMC Health Services Research, 11(1), 1–14. https://doi.org/10.1186/1472-6963-11-207/FIGURES/1

Glass, J. E., McKay, J. R., Gustafson, D. H., Kornfield, R., Rathouz, P. J., McTavish, F. M., Atwood, A. K., Isham, A., Quanbeck, A., & Shah, D. (2017). Treatment seeking as a mechanism of change in a randomized controlled trial of a mobile health intervention to support recovery from alcohol use disorders. Journal of Substance Abuse Treatment, 77, 57–66. https://doi.org/10.1016/j.jsat.2017.03.011

Gliddon, E., Cosgrove, V., Berk, L., Lauder, S., Mohebbi, M., Grimm, D., Dodd, S., Coulson, C., Raju, K., Suppes, T., & Berk, M. (2019). A randomized controlled trial of MoodSwings 2.0: An internet-based self-management program for bipolar disorder. Bipolar Disorders, 21(1), 28–39. https://doi.org/https://doi.org/10.1111/bdi.12669

Graham, A. K., Greene, C. J., Kwasny, M. J., Kaiser, S. M., Lieponis, P., Powell, T., & Mohr, D. C. (2020). Coached Mobile App Platform for the Treatment of Depression and Anxiety Among Primary Care Patients: A Randomized Clinical Trial. JAMA Psychiatry, 77(9), 906–914. https://doi.org/10.1001/jamapsychiatry.2020.1011

Grassi, L., Spiegel, D., & Riba, M. (2017). Advancing psychosocial care in cancer patients. F1000Research, 6. https://doi.org/10.12688/F1000RESEARCH.11902.1

Greenberg, J., Carter, S., Lester, E., Funes, C. J., Macklin, E. A., Plotkin, S., & Vranceanu, A.-M. (2019). Cultivating resiliency in patients with neurofibromatosis 2 who are deafened or

have severe hearing loss: a live-video randomized control trial. Journal of Neuro-Oncology, 145(3), 561–569. https://doi.org/10.1007/s11060-019-03326-5

Heim, E., Ramia, J. A., Hana, R. A., Burchert, S., Carswell, K., Cornelisz, I., Cuijpers, P., el Chammay, R., Noun, P., van Klaveren, C., van Ommeren, M., Zoghbi, E., & Van't Hof, E. (2021). Step-by-step: Feasibility randomised controlled trial of a mobile-based intervention for depression among populations affected by adversity in Lebanon. Internet Interventions, 24, 100380. https://doi.org/10.1016/j.invent.2021.100380

Hensel, J. M., Shaw, J., Ivers, N. M., Desveaux, L., Vigod, S. N., Bouck, Z., Onabajo, N., Agarwal, P., Mukerji, G., Yang, R., Nguyen, M., Jeffs, L., Jamieson, T., & Bhatia, R. S. (2019). Extending access to a web-based mental health intervention: who wants more, what happens to use over time, and is it helpful? Results of a concealed, randomized controlled extension study. BMC Psychiatry, 19(1), 39. https://doi.org/10.1186/s12888-019-2030-x Hensel, J. M., Shaw, J., Ivers, N. M., Desveaux, L., Vigod, S. N., Cohen, A., Onabajo, N., Agarwal, P., Mukerji, G., Yang, R., Nguyen, M., Bouck, Z., Wong, I., Jeffs, L., Jamieson, T., & Bhatia, R. S. (2019). A Web-Based Mental Health Platform for Individuals Seeking Specialized Mental Health Care Services: Multicenter Pragmatic Randomized Controlled Trial. Journal of Medical Internet Research, 21(6), e10838–e10838.

Hesser, H., Gustafsson, T., Lundén, C., & Henrikson, O. (2012). A randomized controlled trial of internet-delivered cognitive behavior therapy and acceptance and commitment therapy in the treatment of tinnitus. Journal of Consulting and Clinical Psychology, 80(4), 649–661. Hoffmann, D., Rask, C. U., Hedman-Lagerlöf, E., Jensen, J. S., & Frostholm, L. (2021). Efficacy of internet-delivered acceptance and commitment therapy for severe health anxiety: results from a randomized, controlled trial. Psychological Medicine, 51(15), 2685–2695. https://doi.org/10.1017/S0033291720001312

https://doi.org/10.2196/10838

Hur, J.-W., Kim, B., Park, D., & Choi, S.-W. (2018). A Scenario-Based Cognitive Behavioral Therapy Mobile App to Reduce Dysfunctional Beliefs in Individuals with Depression: A Randomized Controlled Trial. Telemedicine and E-Health, 24(9), 710–716.

https://doi.org/10.1089/tmj.2017.0214

Jasper, K., Weise, C., Conrad, I., Andersson, G., Hiller, W., & Kleinstäuber, M. (2014). Internet-Based Guided Self-Help versus Group Cognitive Behavioral Therapy for Chronic Tinnitus: A Randomized Controlled Trial. Psychotherapy and Psychosomatics, 83(4), 234–246. https://doi.org/10.1159/000360705

Johansson, P., Westas, M., Andersson, G., Alehagen, U., Broström, A., Jaarsma, T., Mourad, G., & Lundgren, J. (2019). An Internet-Based Cognitive Behavioral Therapy Program Adapted to Patients With Cardiovascular Disease and Depression: Randomized Controlled Trial. JMIR Mental Health, 6(10), e14648–e14648. https://doi.org/10.2196/14648
Kaplan, A. M., & Haenlein, M. (2009). Users of the world, unite! The challenges and opportunities of Social Media. https://doi.org/10.1016/j.bushor.2009.09.003

Kaplan, E. H. (1997). Telepsychotherapy. Psychotherapy by telephone, videotelephone, and computer videoconferencing. The Journal of Psychotherapy Practice and Research, 6(3), 227. /pmc/articles/PMC3330458/?report=abstract

Karabulut, Hatice K et al. Effects of planned group interactions on the social adaptation of individuals with an intestinal stoma: a quantitative study. Journal of clinical nursing vol. 23,19-20 (2014): 2800-13. doi:10.1111/jocn.12541

Kazdin, A. E. (2019). Annual Research Review: Expanding mental health services through novel models of intervention delivery. Journal of Child Psychology and Psychiatry, and Allied Disciplines, 60(4), 455–472. https://doi.org/10.1111/JCPP.12937

Keles, B., McCrae, N., & Grealish, A. (2020). A systematic review: The influence of social media on depression, anxiety and psychological distress in adolescents. International Journal of Adolescence and Youth, 25(1), 79–93. https://doi.org/10.1080/02673843.2019.1590851
Kim, H. J., Kim, S. M., Shin, H., Jang, J.-S., Kim, Y. I., & Han, D. H. (2018). A Mobile Game for Patients With Breast Cancer for Chemotherapy Self-Management and Quality-of-Life Improvement: Randomized Controlled Trial. Journal of Medical Internet Research, 20(10), e273–e273. https://doi.org/10.2196/jmir.9559

Kryger, M. A., Crytzer, T. M., Fairman, A., Quinby, E. J., Karavolis, M., Pramana, G., Setiawan, I. M. A., McKernan, G. P., Parmanto, B., & Dicianno, B. E. (2019). The Effect of the Interactive Mobile Health and Rehabilitation System on Health and Psychosocial Outcomes in Spinal Cord Injury: Randomized Controlled Trial. Journal of Medical Internet Research, 21(8), e14305–e14305. https://doi.org/10.2196/14305

Krzystanek, M., Borkowski, M., Skałacka, K., & Krysta, K. (2019). A telemedicine platform to improve clinical parameters in paranoid schizophrenia patients: Results of a one-year randomized study. Schizophrenia Research, 204, 389–396.

https://doi.org/https://doi.org/10.1016/j.schres.2018.08.016

Lake, J., & Turner, M. S. (2017). Urgent Need for Improved Mental Health Care and a More Collaborative Model of Care. The Permanente Journal, 21, 17–024.

https://doi.org/10.7812/TPP/17-024

Latkin, C. A., Davey-Rothwell, M. A., Knowlton, A. R., Alexander, K. A., Williams, C. T., & Boodram, B. (2013). Social Network Approaches to Recruitment, HIV Prevention, Medical Care, and Medication Adherence. www.jaids.com

Lefebvre C, Glanville J, Briscoe S, Featherstone R, Littlewood A, Marshall C, Metzendorf M-I, Noel-Storr A, Paynter R, Rader T, Thomas J, Wieland LS. Chapter 4: Searching for and selecting studies. In: Higgins JPT, Thomas J, Chandler J, Cumpston M, Li T, Page MJ,

Welch VA (editors). Cochrane Handbook for Systematic Reviews of Interventions version 6.3 (updated February 2022). Cochrane, 2022. Available from www.training.cochrane.org/handbook.

Li, Y., & Tse, M. Y. M. (2020). An Online Pain Education Program for Working Adults: Pilot Randomized Controlled Trial. Journal of Medical Internet Research, 22(1), e15071–e15071. https://doi.org/10.2196/15071

Linden, K., Berg, M., Adolfsson, A., & Sparud-Lundin, C. (2018). Person-centred, web-based support in pregnancy and early motherhood for women with Type 1 diabetes mellitus: a randomized controlled trial. Diabetic Medicine, 35(2), 232–241.

https://doi.org/10.1111/dme.13552

Ljótsson, B., Falk, L., Vesterlund, A. W., Hedman, E., Lindfors, P., Rück, C., Hursti, T., Andréewitch, S., Jansson, L., Lindefors, N., & Andersson, G. (2010). Internet-delivered exposure and mindfulness based therapy for irritable bowel syndrome – A randomized controlled trial. Behaviour Research and Therapy, 48(6), 531–539.

https://doi.org/https://doi.org/10.1016/j.brat.2010.03.003

Lundgren, J. G., Dahlström, Ö., Andersson, G., Jaarsma, T., Kärner Köhler, A., & Johansson, P. (2016). The Effect of Guided Web-Based Cognitive Behavioral Therapy on Patients With Depressive Symptoms and Heart Failure: A Pilot Randomized Controlled Trial. Journal of Medical Internet Research, 18(8), e194–e194. https://doi.org/10.2196/jmir.5556

Ly, K. H., Trüschel, A., Jarl, L., Magnusson, S., Windahl, T., Johansson, R., Carlbring, P., & Andersson, G. (2014). Behavioural activation versus mindfulness-based guided self-help treatment administered through a smartphone application: a randomised controlled trial. BMJ Open, 4(1), e003440–e003440. https://doi.org/10.1136/bmjopen-2013-003440

Ma, C., Wang, B., Zhao, X., Fu, F., Zheng, L., Li, G., & Guo, Q. (2021). WeChat-based education and rehabilitation program in unprotected left main coronary artery disease patients

after coronary artery bypass grafting: an effective approach in reducing anxiety, depression, loss to follow-up, and improving quality of life. Brazilian Journal of Medical and Biological Research = Revista Brasileira de Pesquisas Medicas e Biologicas, 54(4), e10370–e10370. https://doi.org/10.1590/1414-431X202010370

Management of physical health conditions in adults with severe mental disorders WHO GUIDELINES. (2018).

Merckaert, I., Libert, Y., Messin, S., Milani, M., Slachmuylder, J. L., & Razavi, D. (2010). Cancer patients' desire for psychological support: prevalence and implications for screening patients' psychological needs. Psycho-Oncology, 19(2), 141–149.

https://doi.org/10.1002/PON.1568

Merolli, M., Gray, K., & Martin-Sanchez, F. (2013). Health outcomes and related effects of using social media in chronic disease management: A literature review and analysis of affordances. Journal of Biomedical Informatics, 46(6), 957–969.

https://doi.org/10.1016/J.JBI.2013.04.010

Moberg, C., Niles, A., & Beermann, D. (2019). Guided Self-Help Works: Randomized Waitlist Controlled Trial of Pacifica, a Mobile App Integrating Cognitive Behavioral Therapy and Mindfulness for Stress, Anxiety, and Depression. Journal of Medical Internet Research, 21(6), e12556–e12556. https://doi.org/10.2196/12556

Mohr, David C et al. "Behavioral intervention technologies: evidence review and recommendations for future research in mental health." General hospital psychiatry vol. 35,4 (2013): 332-8. doi:10.1016/j.genhosppsych.2013.03.008

Mohr, D. C., Schueller, S. M., Tomasino, K. N., Kaiser, S. M., Alam, N., Karr, C., Vergara, J. L., Gray, E. L., Kwasny, M. J., & Lattie, E. G. (2019). Comparison of the Effects of Coaching and Receipt of App Recommendations on Depression, Anxiety, and Engagement in

the IntelliCare Platform: Factorial Randomized Controlled Trial. Journal of Medical Internet Research, 21(8), e13609–e13609. https://doi.org/10.2196/13609

Muessig, K. E., Nekkanti, M., Bauermeister, J., Bull, S., & Hightow-Weidman, L. B. (2015). A systematic review of recent smartphone, Internet and Web 2.0 interventions to address the HIV continuum of care. Current HIV/AIDS Reports, 12(1), 173–190.

https://doi.org/10.1007/S11904-014-0239-3

Munn, Z., Stern, C., Aromataris, E., Lockwood, C., & Jordan, Z. (2018). What kind of systematic review should i conduct? A proposed typology and guidance for systematic reviewers in the medical and health sciences. BMC Medical Research Methodology, 18(1), 1–9. https://doi.org/10.1186/S12874-017-0468-4/TABLES/1

Naslund, J. A., Aschbrenner, K. A., Marsch, L. A., & Bartels, S. J. (2016). The future of mental health care: peer-to-peer support and social media. Epidemiology and Psychiatric Sciences, 25(2), 113–122. https://doi.org/10.1017/S2045796015001067

Neumayr, C., Voderholzer, U., Tregarthen, J., & Schlegl, S. (2019). Improving aftercare with technology for anorexia nervosa after intensive inpatient treatment: A pilot randomized controlled trial with a therapist-guided smartphone app. International Journal of Eating Disorders, 52(10), 1191–1201. https://doi.org/https://doi.org/10.1002/eat.23152

Nolte, S., Busija, L., Berger, T., Meyer, B., Moritz, S., Rose, M., Schröder, J., Späth-Nellissen, C., & Klein, J. P. (2021). Do sociodemographic variables moderate effects of an internet intervention for mild to moderate depressive symptoms? An exploratory analysis of a randomised controlled trial (EVIDENT) including 1013 participants. BMJ Open, 11(1), e041389–e041389. https://doi.org/10.1136/bmjopen-2020-041389

Obar, J. A., & Wildman, S. S. (2015). Social Media Definition and the Governance Challenge - An Introduction to the Special Issue. SSRN Electronic Journal.

https://doi.org/10.2139/SSRN.2663153

Omidi, Z., Kheirkhah, M., Abolghasemi, J., & Haghighat, S. (2020). Effect of lymphedema self-management group-based education compared with social network-based education on quality of life and fear of cancer recurrence in women with breast cancer: a randomized controlled clinical trial. Quality of Life Research: An International Journal of Quality of Life Aspects of Treatment, Care and Rehabilitation, 29(7), 1789–1800.

https://doi.org/10.1007/s11136-020-02455-z

Oser, M., Wallace, M. L., Solano, F., & Szigethy, E. M. (2019). Guided Digital Cognitive Behavioral Program for Anxiety in Primary Care: Propensity-Matched Controlled Trial.

Osservatorio nazionale sulla salute nelle regioni italiane. (2019). Rapporto Osservasalute sulle cronicità in Italia. https://www.osservatoriosullasalute.it/wp-

JMIR Mental Health, 6(4), e11981–e11981. https://doi.org/10.2196/11981

content/uploads/2019/02/Focus-1-Osservasalute-La-cronicit%C3%A0-in-Italia-feb-2019.pdf Ouzzani, M., Hammady, H., Fedorowicz, Z., & Elmagarmid, A. (2016). Rayyan-a web and mobile app for systematic reviews. Systematic Reviews, 5(1), 1–10.

https://doi.org/10.1186/S13643-016-0384-4/FIGURES/6

Owen, J. E., O'Carroll Bantum, E., Pagano, I. S., & Stanton, A. (2017). Randomized Trial of a Social Networking Intervention for Cancer-Related Distress. Annals of Behavioral Medicine: A Publication of the Society of Behavioral Medicine, 51(5), 661–672. https://doi.org/10.1007/s12160-017-9890-4

Page, M. J., McKenzie, J. E., Bossuyt, P. M., Boutron, I., Hoffmann, T. C., Mulrow, C. D., Shamseer, L., Tetzlaff, J. M., Akl, E. A., Brennan, S. E., Chou, R., Glanville, J., Grimshaw, J. M., Hróbjartsson, A., Lalu, M. M., Li, T., Loder, E. W., Mayo-Wilson, E., McDonald, S., ... Moher, D. (2021). The PRISMA 2020 statement: an updated guideline for reporting systematic reviews. BMJ, 372. https://doi.org/10.1136/BMJ.N71

Pan, D.-N., Hoid, D., Gu, R.-L., & Li, X. (2020). Emotional working memory training reduces rumination and alters the EEG microstate in anxious individuals. NeuroImage. Clinical, 28, 102488. https://doi.org/10.1016/j.nicl.2020.102488

Pendergast, F.J., Ridgers, N.D., Worsley, A. et al. Evaluation of a smartphone food diary application using objectively measured energy expenditure. Int J Behav Nutr Phys Act 14, 30 (2017). https://doi.org/10.1186/s12966-017-0488-9

Perini, S., Titov, N., & Andrews, G. (2009). Clinician-Assisted Internet-Based Treatment is Effective for Depression: Randomized Controlled Trial. Australian & New Zealand Journal of Psychiatry, 43(6), 571–578. https://doi.org/10.1080/00048670902873722

Pew Research Center. (2021, April 7). Demographics of Social Media Users and Adoption in the United States | Pew Research Center. https://www.pewresearch.org/internet/fact-sheet/social-media/?menuItem=81867c91-92ad-45b8-a964-a2a894f873ef

Prescott J, Rathbone AL, Brown G. Online peer to peer support: Qualitative analysis of UK and US open mental health Facebook groups. DIGITAL HEALTH. 2020;6.

doi:10.1177/2055207620979209

Riley L, Cowan M, & Carlos M.C. (2020). Noncommunicable Diseases Progress Monitor 2020. https://www.who.int/publications/i/item/9789240000490

Robinson, E., Titov, N., Andrews, G., McIntyre, K., Schwencke, G., & Solley, K. (2010). Internet treatment for generalized anxiety disorder: a randomized controlled trial comparing clinician vs. technician assistance. PloS One, 5(6), e10942–e10942.

https://doi.org/10.1371/journal.pone.0010942

Robinson-Whelen, S., Hughes, R. B., Taylor, H. B., Markley, R., Vega, J. C., Nosek, T. M., & Nosek, M. A. (2020). Promoting psychological health in women with SCI: Development of an online self-esteem intervention. Disability and Health Journal, 13(2), 100867. https://doi.org/https://doi.org/10.1016/j.dhjo.2019.100867

Ronen, K., Grant, E., Copley, C., Batista, T., & Guthrie, B. L. (2020). Peer Group Focused eHealth Strategies to Promote HIV Prevention, Testing, and Care Engagement. Current HIV/AIDS Reports, 17(5), 557–576. https://doi.org/10.1007/S11904-020-00527-W Ruehlman, L. S., Karoly, P., & Enders, C. (2012). A randomized controlled evaluation of an online chronic pain self management program. Pain, 153(2), 319–330.

https://doi.org/10.1016/j.pain.2011.10.025

Ruland, C. M., Andersen, T., Jeneson, A., Moore, S., Grimsbø, G. H., Børøsund, E., & Ellison, M. C. (2013). Effects of an Internet Support System to Assist Cancer Patients in Reducing Symptom Distress: A Randomized Controlled Trial. Cancer Nursing, 36(1). https://journals.lww.com/cancernursingonline/Fulltext/2013/01000/Effects_of_an_Internet_S upport System to Assist.4.aspx

Rupert DJ, Gard Read J, Amoozegar JB, et al. Peer-Generated Health Information: The Role of Online Communities in Patient and Caregiver Health Decisions. Journal of Health Communication. 2016 Nov;21(11):1187-1197. DOI: 10.1080/10810730.2016.1237592. PMID: 27805496; PMCID: PMC7315436.

Schlosser, D. A., Campellone, T. R., Truong, B., Etter, K., Vergani, S., Komaiko, K., & Vinogradov, S. (2018). Efficacy of PRIME, a Mobile App Intervention Designed to Improve Motivation in Young People With Schizophrenia. Schizophrenia Bulletin, 44(5), 1010–1020. https://doi.org/10.1093/schbul/sby078

Schulz, S. M., Ritter, O., Zniva, R., Nordbeck, P., Wacker, C., Jack, M., Groschup, G., Deneke, T., Puppe, F., Ertl, G., Angermann, C., Störk, S., & Pauli, P. (2020). Efficacy of a web-based intervention for improving psychosocial well-being in patients with implantable cardioverter-defibrillators: the randomized controlled ICD-FORUM trial. European Heart Journal, 41(11), 1203–1211. https://doi.org/10.1093/eurheartj/ehz134

Scogin, F., Lichstein, K., DiNapoli, E. A., Woosley, J., Thomas, S. J., Larocca, M. A., Byers, H. D., Mieskowski, L., Parker, C. P., Yang, X., Parton, J., McFadden, A., & Geyer, J. D. (2018). Effects of Integrated Telehealth-Delivered Cognitive-Behavioral Therapy for Depression and Insomnia in Rural Older Adults. Journal of Psychotherapy Integration, 28(3), 292–309. https://doi.org/10.1037/INT0000121

Smailhodzic, E., Hooijsma, W., Boonstra, A. *et al.* Social media use in healthcare: A systematic review of effects on patients and on their relationship with healthcare professionals. BMC Health Serv Res 16, 442 (2016). https://doi.org/10.1186/s12913-016-1691-0

Smith, A., & Anderson, M. (2018, March 1). Social Media Use in 2018.

https://www.pewresearch.org/internet/2018/03/01/social-media-use-in-2018/

Sokol, R., & Fisher, E. (2016). Peer support for the hardly reached: A systematic review.

American Journal of Public Health, 106(7), e1-e8.

https://doi.org/10.2105/AJPH.2016.303180

Starri, M. (2021, January 27). Digital 2021: i dati globali.

https://wearesocial.com/it/blog/2021/01/digital-2021-i-dati-globali/

Stein, D. J., Benjet, C., Gureje, O., Lund, C., Scott, K. M., Poznyak, V., & van Ommeren, M. (2019). Integrating mental health with other non-communicable diseases. BMJ, 364.

https://doi.org/10.1136/BMJ.L295Stellefson M, Chaney B, Barry AE, Chavarria E, Tennant B, Walsh-Childers K, Sriram P, Zagora J

Web 2.0 Chronic Disease Self-Management for Older Adults: A Systematic Review J Med Internet Res 2013;15(2):e35 doi: 10.2196/jmir.2439

Stinson, J., Ahola Kohut, S., Forgeron, P., Amaria, K., Bell, M., Kaufman, M., Luca, N., Luca, S., Harris, L., Victor, C., & Spiegel, L. (2016). The iPeer2Peer Program: a pilot

randomized controlled trial in adolescents with Juvenile Idiopathic Arthritis. Pediatric Rheumatology Online Journal, 14(1), 48. https://doi.org/10.1186/s12969-016-0108-2 Strandskov, S. W., Ghaderi, A., Andersson, H., Parmskog, N., Hjort, E., Wärn, A. S., Jannert, M., & Andersson, G. (2017). Effects of Tailored and ACT-Influenced Internet-Based CBT for Eating Disorders and the Relation Between Knowledge Acquisition and Outcome: A Randomized Controlled Trial. Behavior Therapy, 48(5), 624–637.

https://doi.org/https://doi.org/10.1016/j.beth.2017.02.002

Titov, N., Andrews, G., Choi, I., Schwencke, G., & Mahoney, A. (2008). Shyness 3:
Randomized Controlled Trial of Guided Versus Unguided Internet-Based CBT for Social
Phobia. Australian & New Zealand Journal of Psychiatry, 42(12), 1030–1040.
https://doi.org/10.1080/00048670802512107

Titov, N., Andrews, G., Davies, M., McIntyre, K., Robinson, E., & Solley, K. (2010). Internet treatment for depression: a randomized controlled trial comparing clinician vs. technician assistance. PloS One, 5(6), e10939–e10939. https://doi.org/10.1371/journal.pone.0010939

Titov, N., Andrews, G., Johnston, L., Robinson, E., & Spence, J. (2010). Transdiagnostic Internet treatment for anxiety disorders: A randomized controlled trial. Behaviour Research and Therapy, 48(9), 890–899. https://doi.org/https://doi.org/10.1016/j.brat.2010.05.014

Titov, N., Andrews, G., Robinson, E., Schwencke, G., Johnston, L., Solley, K., & Choi, I. (2009). Clinician-Assisted Internet-Based Treatment is Effective for Generalized Anxiety Disorder: Randomized Controlled Trial. Australian & New Zealand Journal of Psychiatry, 43(10), 905–912. https://doi.org/10.1080/00048670903179269

Titov, N., Andrews, G., Schwencke, G., Drobny, J., & Einstein, D. (2008). Shyness 1: Distance Treatment of Social Phobia Over the Internet. Australian & New Zealand Journal of Psychiatry, 42(7), 585–594. https://doi.org/10.1080/00048670802119762

Titov, N., Dear, B. F., Schwencke, G., Andrews, G., Johnston, L., Craske, M. G., & McEvoy, P. (2011). Transdiagnostic internet treatment for anxiety and depression: A randomised controlled trial. Behaviour Research and Therapy, 49(8), 441–452.

https://doi.org/https://doi.org/10.1016/j.brat.2011.03.007

Topooco, N., Byléhn, S., Dahlström Nysäter, E., Holmlund, J., Lindegaard, J., Johansson, S., Åberg, L., Bergman Nordgren, L., Zetterqvist, M., & Andersson, G. (2019). Evaluating the Efficacy of Internet-Delivered Cognitive Behavioral Therapy Blended With Synchronous Chat Sessions to Treat Adolescent Depression: Randomized Controlled Trial. J Med Internet Res, 21(11), e13393. https://doi.org/10.2196/13393

Van Daele, T., Karekla, M., Kassianos, A. P., Compare, A., Haddouk, L., Salgado, J., Ebert, D. D., Trebbi, G., Bernaerts, S., Van Assche, E., & De Witte, N. A. J. (2020).

Recommendations for policy and practice of telepsychotherapy and e-mental health in Europe and beyond. Journal of Psychotherapy Integration, 30(2), 160–173.

https://doi.org/10.1037/int0000218

Vilhauer, R. P., McClintock, M. K., & Matthews, A. K. (2010). Online Support Groups for Women with Metastatic Breast Cancer: A Feasibility Pilot Study. Journal of Psychosocial Oncology, 28(5), 560–586. https://doi.org/10.1080/07347332.2010.504504

Voinov, B., Richie, W. D., & Bailey, R. K. (2013). Depression and Chronic Diseases: It Is Time for a Synergistic Mental Health and Primary Care Approach. The Primary Care Companion for CNS Disorders, 15(2), 26226. https://doi.org/10.4088/PCC.12R01468

Wang, Q.-Q., Zhao, J., Huo, X.-R., Wu, L., Yang, L.-F., Li, J.-Y., & Wang, J. (2018). Effects of a home care mobile app on the outcomes of discharged patients with a stoma: A randomised controlled trial. Journal of Clinical Nursing, 27(19–20), 3592–3602. https://doi.org/10.1111/jocn.14515

Weise, C., Kaiser, G., Janda, C., Kues, J. N., Andersson, G., Strahler, J., & Kleinstäuber, M. (2019). Internet-Based Cognitive-Behavioural Intervention for Women with Premenstrual Dysphoric Disorder: A Randomized Controlled Trial. Psychotherapy and Psychosomatics, 88(1), 16–29. https://doi.org/10.1159/000496237

Weise, C., Kleinstäuber, M., & Andersson, G. (2016). Internet-Delivered Cognitive-Behavior Therapy for Tinnitus: A Randomized Controlled Trial. Psychosomatic Medicine, 78(4). https://journals.lww.com/psychosomaticmedicine/Fulltext/2016/05000/Internet_Delivered_C ognitive_Behavior_Therapy_for.13.aspx

Wims, E., Titov, N., Andrews, G., & Choi, I. (2010). Clinician-Assisted Internet-Based
Treatment is Effective for Panic: A Randomized Controlled Trial. Australian & New Zealand
Journal of Psychiatry, 44(7), 599–607. https://doi.org/10.3109/00048671003614171
Wise, M., Marchand, L. R., Roberts, L. J., & Chih, M.-Y. (2018). Suffering in Advanced
Cancer: A Randomized Control Trial of a Narrative Intervention. Journal of Palliative
Medicine, 21(2), 200–207. https://doi.org/10.1089/jpm.2017.0007

Wootten, A. C., Abbott, J.-A. M., Meyer, D., Chisholm, K., Austin, D. W., Klein, B., McCabe, M., Murphy, D. G., & Costello, A. J. (2015). Preliminary Results of a Randomised Controlled Trial of an Online Psychological Intervention to Reduce Distress in Men Treated for Localised Prostate Cancer. European Urology, 68(3), 471–479.

https://doi.org/https://doi.org/10.1016/j.eururo.2014.10.024

Wootten, A. C., Meyer, D., Abbott, J.-A. M., Chisholm, K., Austin, D. W., Klein, B., McCabe, M., Murphy, D. G., & Costello, A. J. (2017). An online psychological intervention can improve the sexual satisfaction of men following treatment for localized prostate cancer: outcomes of a Randomised Controlled Trial evaluating My Road Ahead. Psycho-Oncology, 26(7), 975–981. https://doi.org/https://doi.org/10.1002/pon.4244

World Health Organization. Noncommunicable Diseases and Mental Health Cluster. (2002). Innovative care for chronic conditions: building blocks for actions: global report.

https://apps.who.int/iris/handle/10665/42500

Young, S. D., Koussa, M., Lee, S.-J., Perez, H., Gill, N., Gelberg, L., & Heinzerling, K. (2018). Feasibility of a social media/online community support group intervention among chronic pain patients on opioid therapy. Journal of Addictive Diseases, 37(1–2), 96–101. https://doi.org/10.1080/10550887.2018.1557992

Zwerenz, R., Becker, J., Gerzymisch, K., Siepmann, M., Holme, M., Kiwus, U., Spörl-Dönch, S., & Beutel, M. E. (2017). Evaluation of a transdiagnostic psychodynamic online intervention to support return to work: A randomized controlled trial. PloS One, 12(5), e0176513–e0176513. https://doi.org/10.1371/journal.pone.0176513

Table 1. PICOS approach

Table 2. Characteristics of the interventions

Figure 1. Study selection process